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SUMMARY OF THE WEEK
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Last news from yesterday 16:00

Target polarization with data at low intensity for Trigger efficiency
20:00 DY started (PAO3V pb) '

1:45 CO2 GASALARM - GEM, MWPC, STRAW, W45, MW1, RW HV OFF

At 1:45 we observed CO2 flow drop on GEMs and MWPCs. Soon after we had to switch off the HVs. We called
Vincent, who tried to check the PLC, but didn't find a solution. Then he exchanged many calls with TCR and various
piquets trying to find who is responsible for our gas supply. At 3:15 the problems appeared also on STRAW gas
system and at 3:40 on W45 (they are recycling the gas, as well as MW1, where we see no problems so far).

4:00 Vincent and Bogdan readjusted the PLCs to restore Ar flow in GEM, MWPC, STRAW and W45. The Ar flow
stopped because the supply of CO, from CERN had stopped. After ~1 h without flushing, we started to see a
humidity alarm on STRAW, which disappeared after the Ar flow was restored. However, we still have no CO,.

8:00 Beam Stop for Main power converters intervention until 12:00

8:42 First the technical service (72201) and then the responsible for NA, David Jaillet (167151) told us that the
intervention on CO2 supply is finished.

The problem is the same as in the past, namely a not properly working switch from an empty bottle to a new one.

- 12:00 Hoping Beam back with operational detectors
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The situation resembles the incident from 12. 6. 2017, ~7:00, when the CO2 supply to COMPASS was
stopped due to an abnormally high consumption elsewhere in North area (leak). The corresponding
comment is referenced.



Period: 2018-10-12 10:59:59 - 2018-10-19 11:00:00
30000

27758 SPS eff = N3/N1=90%

25233

25000 COMPASS DY eff = N5/N3= 80%

21225

20000

15000

s

10000

5000

SPS SPS SP5 COMPASS Drell-Yan Run type
SFTPRO FTARGET TG == 110.0 TE == 110.0 T6 == 110.0



T6 intensity/E11 (protons) integrated
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The Beam

Mainly 2 spills in the favorite SC of 28.8 s
but sometimes
2 spillsin 32.4s, 36s, 43.2s, 44 .4s,
46.8s, 48.6s, 50.4s... 80.4s

SPS-PAGE1 Current user: AWAKE1 2.77E+11 16-10-18 22:21:43
SC 10 (42BP, 50.4s) Last update: 1 seconds ago

Tuesday evening
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| Strang 50Hz component

92 a HE/H8
95 a COMPASS

P Trhdety Dhp  PPo FS dric b iy o 2

bt el licaroiit 47 NAB2
: Phone: 77500 or 70475
Frequency (HZ) SFTPROZ 2892 E10 2790 E10 Comments (16-Oct-2018 16:09:47)

SPS-PAGE1 Current user: LHCION1 1.11E+11 16-10-18 10:10:25

SC 5 (67BP, 80.4s)

DDESTECO

I/E11
20.1
38.9
124.7
21.9

Last update: 41 seconds ago

Tuesday morning

Experiment
H2/H4
HE6/H8
COMPASS
NAG2
Phone: 77500 or 70475
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Target Polarization
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A few detector pbs



MWPC PAO4 ID453

Has been Repaired on Wednesday 17 Oct before 16:00
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Thursday 22:00 Alexander port 0 excluded
M W PC PAO3 I D45 3 Friday 6:00 Bogdan port 0 back in FE database
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MW1 noisy channels (disappear and reappear)
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channel#

RICH Crate has been exchanged

RUN 286962 Thursday 18 Oct, 23:21
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MM — Remarks For shift Crew

Noisy channels:
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DCOO0 and DCO1 — Remarks For shift Crew
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2 stations of 2 perpendicular wires added
to the proton radlus measurement setup
Preparation of th gl AR - see Christian’s Talk
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Survey was done




The situation day by day
is presented in the spare slides



Thanks to the shift crews
and to the detector and target experts

A great thank to Vincent. Michael. Christophe,
Annika, Moritz, Johannes...

Good Luck to Robert!
(Take care of the manip but also of your health)






A short sumary from Friday 12 October 12:00

v From Friday 12:00 to Saturday 8:00 - pb on the beam 9.5 hours lost on 20hours
v From Saturday 8:00 until Sunday 16:00 — good beam and good data taking

v From Saturday 16:00 until Monday 16:00 — one pb on the beam (1h) and a few detector pb

As usual mainly 2 spills in the favorite SC of 28.8 s

but sometimes 2 spills in 32.4s, 36s, 43.2s, 44.4s, 46.8s, 48.6s, 50.4s... 80.4s
and 1 spill in 37.2s during LHC filling
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From Saturday 13 Oct 8:00 to Sunday 14 Oct 16:00
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From Sunday 14 Oct 16:00 to Monday 15 Oct 16:00
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A few detector problems



MWPC PAO4 |ID453

On Monday night and morning MWPC PAO4 ID 453 persisting errors
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This was fixed by Bogdan Vasilishin during an access this morning
One card lost LV connection)

And several reloads

And after 13:00 again the pb



MMO1 ID 380

Rainer’s comment: noisy MMO01 380 but no errors in MurphyTV

Rainer’s recipe: As reloading (LOAD -ra 380) did not fix the problem,

we power-cycled the LV.

After that LOAD -ra 380 (stated on the whiteboard) did not solve the problem,
however, LOAD -zRCpdrat 380 (given in the instruction folder) did.

Damien’s recipe: LOAD —A 380 and (LV power cycle in case)

MFGT LA az

PO LA oocupancies ; P _
[ PO LA hithAultiplicity | ]
B 10 EnirsedT 5
1 Eniriedsz234 E
- F fdean 1om.3
naz !#‘ 1o blean £3.54 N = R
. F5 o7l
o' b
oia
1%
o
016 ) W ,
o’ b . i
014
oz
ol 1‘!"{:‘5_- 10 E_
n.og ne 'JI:.':I I -
- : %
- S L] E
oo M\ 3 f|I‘l: I -
"y
ooz | 1 ' 1E
E B III IIIIIIIIIIIIII

|||I||||I||||I||
a oo 0 100 E00 300 400 SO0 SO0 FOO BOO 500

a 200 40 [alu] ] 1000 120 a = ioo iso ] 350




MW?2ID 416

62297 R Joosten pof 12 October 2018, w2 MW2 ok again

Froblem in the end solved by reloading very often

Refers to #62296
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MW1 noisy channels
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Straw HV trips

62301

R. Joosten

Maon 15 October 2018,
07:02

STRAW

Trip in #U1_DLO6 and 3Y2_DLO1

HV SYSTEM

B

(91.DLos 8Y1_DLo7 U1 DLos.

aViDlez ava.Dlen axeDios.

Straw 03U1 DoubleLayer D&:

Group aperalon.
On o

Fop Window
AgzlEan

Camrsl Hams
E_Hv_Di0E_adi_Emm_0_FH
El_Hv_D0e paU1_Bam_1_Baeve
£1_Hv D8 AU _Baim 2

H1_Hw (308 (6L1_Bnm_3
S_Hv_DH DRU1_Bam_4_Ph
S_Hv_DOG_Lad1_fnm_b
B_Hv_DOE_DaJ1_Enm_&
E_Hy_D0E LAl 1_Emm_7_Jum
Ei_Hy_D0E 01 _10mm_Sakes

Straw D3Y2 DoubleLayer 01:

Group oparation:
Cn oH

Pap Windaw
Ariion

Chanisl lens
=_Hy_DO_OFE_tmm P
S_H_DD1_0R2_Emm_1_BOTTOM
S_Ha_DE_0PYD_Bmm_j
F_Hv_DO7_DF2_fmm_3_Pr

S He_DD1_OF2_Emm_4_P=

St _DEN_{xFr_rm_§

= _Hv_DD1_OF2_Emm_E_TOP
[FEEET ]

Matusacl!
S_Hv_DOA_OF2_1omm_BOTTOM
2_Hs_ DO _00E 10mn_1

I _DO_0FvE_1omn_2

Lo ARELE]
1540 D0
TEA0 (X0
11540 D0
1540 000
1R (XD
1540 D0
1E40 B0
L AR E]
1 70, oo
1 750 (0
1 70

L AR44 K]
=40 00
B0 00
0
1 70, D0
1 Tl
L]

Pobonii A
[0

5

4 ¥R

atan
€100
L]

idax iy
1540 000
T
T
T XN
1] [T
14540 000
1540 000
1540 000
1780 000

Wiina [y
I
150,000
B0
1BI000
15400000
TR
1640,000
TR0
LELTEY
178,000
1TEN
1TH00G

switched HV back on and full HV is reached again

L e
20 000
20000
30 000
110
1 X
20 0
20000

A
b ele vl
el nil
A
200000
LR h ]
20000

(=11 Heakarms
&R —wAlir T

62295

R. Joosten

Mon 15 October 2018,
02:22

STRAW

HV SYSTEM TEMPERATURES
(Ba (SMBEE avipter sUimies | o [sms | | [spsnl | | O | [Gesuen
Vi Dlez sY2_Dier  3x2_DLos
01 Doublelayer O6:
wralion:  Channel Hams WOV vk [ = [ TN WA LENTTEN] Eon Hwakrms
oF 5 Hw_DOE_Ced_Emm_0_PH 1640 000 240 100 oo 40000 SO0

B1_Hw_DOE_DEd1_Bmm_1_Sakws 1640 [0 140 OO0 008 EHION EO00
Fi_Hw_DOE_Dad1_Bmm_2 ATk DR D00 0 ) VR0 000 20000

rdow SI_rv_DOE_CeE1_Bmm_3 LR R o] a0 100 101 0eX) R0 000 20000

an Si_Hv_DOE_DEx1_Bmm_4_Ph i, DO i aln b i) [RRE) [4] T D 2
B_Hv_DOE_OE61_Bmm_5 1640000 NS0 0D (el e af &0 00 E0m
S1_Hv_DOE_OEK1_Smm_B 16540000 HE40. 100 0060 E40000 E0D00
S1_Hv_DIOE_DE1_Smm_7_Jum 160 00 140 000 010G EH0000 E000
S Hw_OE_M_t0m e _Saie 17E0 00X + TR0 000 o TRIOOD 3000
FI_Hw_DOE_DEx1_1dmm_1 170000 T TR0 000 71X TR0 000 200000
Si_He_DOE_DE1_10mm_2 170000 Al k] TE40 1 TR0 000 2
B_Hw_DOE_DEN1_1omm_2 1780000 o [en] B0 il i n ] 0
S1_Hv DOE DEx1_10mm_4 1 7EDLDOO 0000 17T8I.000 pralan ]

switched on again; it reaches full HY without tripping




DCS: Polarized target
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Trap Nitrogen level

Trending Plot X
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From Monday 15 Oct 16:00 to Tuesday 16 Oct 16:00

Time Range = | ¥ Axes = | Other = |1:1 log auto scientific = Cloze
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Last problem just now from 15:17 to 15:37 Gas problem

(Ar line switched to another line on battery-

we will be back on the normal line on Friday — impurity 02, H20...too large fracton of C2H6 or not’

Drop of Ar flow for a couple of minutes during the rerouting of the Ar flow.
PLC was reset for DCs as it was in alarm.

DETECTOR CONTROL SYSTEM - 52195 Tuesday 16102018
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A few problems

v" DAQ: IP-BUS network switch hang - Power cycle of the switch solved the problem (Vladimir F.)

v' DCO5 100% errors - Reload IDs 882 883 884 885 886 887 888 889
v/ DCO1 - ID 257 port 5 - 100% errors - Reload

v MWPC PAO4 ID 453 — one time the missing channels were back after an ultilate reload attempt.... But only for 30mn
v MWPC PAO5 ID 452 - reload successful

v' RW ID 432 errors — a reload is only necessary at the end of the run

PAO4UT_ ch | PAQAL1___ch

PAO4L1_ ch

PAD4UT_ dh
| Entries 16002 12
140 __ i Mean 344 ‘;
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RAMS 170.5

RMS 155
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A few problems

v" DCS CAEN manager

DPC Managers & Hardware far CAEN
Hardware controlled by CAEN OPC Server
Name State Heart beat Action r_JEJn:-};;Z;__""""""""""'": R e ———
] []
- scem | 11 .
ISEG HW . 4| 2018.10.16 02:36:45.638 Start | Stop |1 .
1 1 — 1 Ly :
— — ¥ :
CAEN 1 HW . 4| 2018.10.16 02:25:55.110 Start | Stop -- i E- :
r . 1 ! (]
CAEN 2 HW . & | 20181016 02:37:10.848 T e R B | | E—
SCHNEIDER |Hw @ x| 20181016 02:36:37.:380 Start | Stop N R —— Etheme
I s 18-
CANOPEN  |Hw . 4| 2018.10.16 02:37:01.285 Start | Stop == w___
1 HOD ok Culer 1Y, =oce veio HY
: HOD_CCRT Haooh Weio HY
WIENER1  [nw| | (@ g [(2018.10.16 02:36:49.101 Start | Stop | oo e s e
1 RIGH_CE: A 1 HY MAFHTS Top H1 HY
[ Raa I_'JHI'_I.::I-" HWiL Ve
WIENER2  |Hw . 4 | 20181016 02:36:47 349 Start | Stop i
WIENERZ  |Hw/ . 4 ||2018.10.16 02:36:44.502 Start | Stop Clove

1- During the night lost of communication with RICH_CCR2 and _CCR3 (RICH HV MAPMTs + H1)
Stop/Start sloved the problem
2- This morning only lost of communication with RICH_CCR2 (RICH HV MAPMTs Bottom) .
An access for Christophe to see that the HV crate was not running. Powercycle did not help.
=» The alarm is masked and the CAEN crate will be repaired when we have no beam (on Wednesday or Thursday)

v DCS ECAL1 position timestamp stopped. Restart of pccoef23 by Sergei D. during the access this morning.



Tue 16 October 2018,

1 - J. Matousek 17-09

RICH Bottom half of RICH is missing, because of the unresponsive cr
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TO DO LIST FOR MD
on Wednesday 17 October from 8:00 to 18:00
v" GEM OFF and SM1+SM2 OFF

v’ Target Polarization

v' Na58pi006 not responding after power cycling?

v MWPC PAO4 ID 453 (just at the moment maybe the pb is cured?!)
v’ Installation of SciFis for Proton Radius (Munich Team)

v RICH HV CAEN crate for MAPMT Bottom part

v’ Visit of the COMPASS hall by a few Czech VIP with Michael

v CEDARS activity when the beam is back at 18:00






From Tuesday 16 Oct 16:00 to Wednesday 17 Oct 16:00

Time Range = | ¥ Axes = | Other = |11 Iog) auto scientific 3 Cloze
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many pbs on detectors during Beam with a strong 50Hz component

Run number: 286917 21:37 o —
SourcelD 453: (PAO3V1__ PA04U1__ PA04V1__ PA04X1_ )

1047 (100% bad events) Sum of errors on this SourcelD
1047 (100% bad events) DAQ ERROR: missing sourcelD(s)

SourcelD 380: (MPOTMU__ MPO1MV__ MPO1MX__ MPO1MY_ MP01U1__ MP0O1V1__ MPO1X1__ MPO1Y1_ )

535 (51.1% bad events) Sum of errors on this SourcelD
535 (51.1% bad events) DAQ ERROR: missing sourcelD(s)

SourcelD 454: (PA05U1__ PAO5V1__ PAO5X1 )

465 (44.4% bad events) Sum of errors on this SourcelD
465 (44.4% bad events) DAQ ERROR: missing sourcelD(s) abs value of FFT // Tue Oct 16 21:13:45 2018

SourcelD 452: (PA02U1__ PA02V1__ PAO3U1__ PAO3X1_ )

465 (44.4% bad events) Sum of errors on this SourcelD B StrongSOHzcomponent ______________________________________________________________ —
465 (44 .4% bad events) DAQ ERROR: missing sourcelD(s) 40000 [ ........................................... I .................................................................. .....................

SourcelD 451: (PAO1U1_ PAOTV1__ PAO1X1__ PAO2X1__ )
465 (44.4% bad events) Sum of errors on this SourcelD

465 (44.4% bad events) DAQ ERROR: missing sourcelD(s) S R S S R
frequency [Hz]

| Tue 16 October 2018,
122:17

Error from port 5 propageted to SrcID 452, 451, 454 so | excluded it from DB. At this moment two ports are missing in PA04X1. Tomorrow during MD
I will try to fix them.

62324 | B. Vasilishin MWPC | PA04X1 (SrclD 453, port5, port7)




Difficult data taking on yesterday evening

MLAST, OLAST

286022 | 2018~ Tue 16 October 2018, 01:10 1200 |drell-yan | Good | Prob. CT, LASLAST, DY - 100% errors on MWPC(453) and RW(432)
W19 | 23:58
TRand
MLAST, OLAST,
286921 | 2018- | Tue 16 October 2013, 23:38 6 drell.yan | Good | Good CT, LASLAST, DY
W19 | 23:35
TRand
MLAST, OLAST,
286920 | 2018-  Tue 16 October 2018, 23:33 4 drell.yan | Good | Good CT, LASLAST, DY
W19 | 23:31
TRand
MLAST, OLAST,
ol F=Eell S S 23:31 200 |drell-yan | Good | Good CT, LASLAST, DY
W19 | 22:08
TRand
286018 2018-  Tue 16 October 2018, 5444 4 drell-yan  Prob. | Prob. DY, PAO4 missing
W19 21:41
MLAST, OLAST,
286017 2018~ Tue 16 October 2018, 21:38 5 drell-yan | Prob.  Prob. CT, LASLAST, DY, PAO4 missing, MPO1MY problem
W19 21:36
TRand
MLAST, OLAST,
286016 2018~  Tue 16 October 2018, 21:18 4 drell-yan | Good  Good CT, LASLAST, DY, strong 50 Hz in the beam
W19 21:17
TRand
MLAST, OLAST,
286915 53113' ;‘?0;6 October 2018, 21:12 11 drel-yan Good Good CT, LASLAST, DY, strong 50 Hz in the beam

TRand




During MD on Wednesday 17 October from 8:00 to 18:00

v' GEM OFF and SM1+SM2 OFF
v’ Beam line ON as tests for communication bus problem

v’ Visit of the COMPASS hall by a few Czech VIP with Michael

v’ Target Polarization started at 8:30
v’ Status of Na58pi006?

v' Installation of SciFis for Proton Radius (Munich Team)

v MWPC PAO4 ID 453 + RW ID 432

v RICH HV CAEN crate for MAPMT Bottom part

v’ ECAL2 LV

v' DAQ new multiplexer installed (First 4 ports of MUX01 connected to xSwitch)

v Close the doors 211 & 231 v’ Safety visit for the scintillator target

v' CEDARS activity when the beam is back at 18:00 for the proton radius measurement
v’ Japoneese VIP visit with Nori



During the previous polarization

Today
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Box with the 2 staﬁns}
with 2 perpendicular wires
Survey was done







From Wednesday 17 Oct 16:00 to Thursday 17 Oct 16:00

Tim= Range —~ | %Y Awxes | Other —  1:1 o auto scientific 3 Cloze
N
= ]
T 8- ’H
25—
T 18:30 = 23:3C
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Only very few problems on detectors during the night

At 8:00
At 1.50 MPOZUI_netulipicity ___ MPOVT__ hithustiglicity :

DCO1V1__ ch Run999992 e
120:—
IIJO:—
i and
EI}:— °

: RichWall errors ID432
::; MPOZMU_ hebuttiplicity _— MPO2MY  hiadultipicity _— R e a p p e a r e d
N 1 R T T

LV powercycle both on ASD8 and F1




During the previous polarization

Yesterday and Today

Time Range = | Y Axes = | Other =|1:1 g auto =cientific 1 Time Range = | ¥ Axes = | Other = |11 log auto scientific
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Today, Thursday 18 October, NO BEAM from 8:00 to 16:00

v' GEM OFF and SM1+SM2 OFF
v' Beam line ON as tests for communication bus problem

v’ Target Polarization started at 8:30
v Reparation of Na58pi006°?

v Improvement of the setup with of SciFis for Proton Radius (Munich Team)

v RW ID 432
v RICH HV CAEN crate for MAPMT Bottom part

At 14:15
v’ Safety Visit for the scintillator target for the proton radius measurement
v’ Visit of the COMPASS hall by a few Japoneese VIP with Nori

v’ Data taking with low intensity (Trigger efficiencies, CEDARs test, ...)
until the polarization reaches its maximum






[dcs@pccompass®7 W42]1$ more summary 2018-10-12 10\:59\:59 2018-10-19 11\:00\:00.txt
Summary

start time: 2018-10-12 10:59:59
end time: 2018-10-19 11:00:00

#1 SFTPRO attempts: 27758

#2 Number of spills marked as FTARGET: 25664

#3 Number of spills marked as FTARGET and T6 == 110.0: 25233
#4 Number of COMPASS spills and T6 >= 110.60: 21225

#5 Number of DY Run type spills and T6 == 110.0: 20159

Ratios:

#2 / #1 = 0.92
#3 / #2 = 0.98
#4 / #3 = 0.84
#5 / #3 = 0.80

Mean values:

SPS T6 >= 110.0 mean value: 121.72

SPS ion2 (raw) mean value for FTARGET T6 == 110.0: 55633.72

SPS ion2 (calibrated) mean value for FTARGET T6 == 110.08: 2.9e+08
COMPASS T6 >= 110.0 mean value: 121.78

COMPASS ion2 (raw) mean value for FTARGET T6 == 110.0: 62015.82
COMPASS ionZ (calibrated) mean value for FTARGET T6 == 110.0: 3.3e+08
COMPASS DAQ dead time mean value: 13.43

COMPASS Veto dead time (Random) mean value: 12.53

COMPASS Veto dead time (MT) mean value: 20.36

COMPASS Veto dead time (0T) mean value: 18.03

DY Run type T6 == 110.0 mean value: 121.75

DY Run type ion2 (raw) mean value for FTARGET T6 == 110.0: 64991.53
DY Run type ion2 (calibrated) mean value for FTARGET T6 == 110.0: 3.4e+08



